Author's response to reviews

Title: Inactivation and sub-lethal injury of Salmonella Typhi, Salmonella Typhimurium and Vibrio cholerae in copper water storage vessels

Authors:

   Riti Sharan (riti.sharan@gmail.com)
   Sanjay Chhibber (sanjaychhibber8@sify.com)
   Robert H Reed (r.reed@cqu.edu.au)

Version: 4 Date: 22 July 2011

Author's response to reviews:

Dear Editor,

The following changes have been made in the manuscript as requested by the production team:

1. The range used in listing of figures has now been changed on page 11.
2. The line numbers have been removed.
3. The highlighting has been removed from the text.
4. The term “et al.” has been removed from the reference list and all the authors have now been listed.
5. Typographic errors have been checked.

Please don’t hesitate to contact the authors in case of any further assistance.

Thanks & Regards

Dr. Riti Sharan